Author's response to reviews

Title: Prescription of respiratory medication without an asthma diagnosis in children: a population based study.

Authors:

Mira GP Zuidgeest (m.zuidgeest@uu.nl)
Liset van Dijk (l.vandijk@nivel.nl)
Henriette A Smit (jet.smit@rivm.nl)
Johannes C van der Wouden (j.vanderwouden@erasmusmc.nl)
Bert Brunekreef (b.brunekreef@uu.nl)
Hubert GM Leufkens (secrfenf@pharm.uu.nl)
Madelon Bracke (m.bracke@uu.nl)

Version: 5 Date: 15 January 2008

Author's response to reviews:

January 15, 2008
To the Central Editorial Production Team

BMC Health Services Research


L.S.,

We have amended the manuscript in line with the requested formatting changes.

The abstract in the submission system has been updated to exactly match the abstract in the manuscript.

An Acknowledgements section has been added to the manuscript.

The appearance of the tables have been adjusted.

Table 4 and 5 are intended to be additional files, who do not need to be visible in the manuscript, so no changes are made there.

Yours sincerely,

Mira G.P. Zuidgeest on behalf of all co-authors